Author's response to reviews

Title: A case of a perforated obturator hernia with a femoral abscess repaired using the pectineus muscle

Authors:

Naoki Kubo (nkazumihp@yahoo.co.jp)
Zyunichi Yoshizawa (ciel001100@gmail.com)
Takaomi Hanaoka (Ghanaoka@gmail.com)
Koichi Nakamura (k_nakarc5@yahoo.co.jp)

Version: 3 Date: 16 February 2015

Author's response to reviews: see over
Case Report Cover Letter

2/11/2015

Dr. Guangde Tu
Editor    BMC Surgery

Dear Dr. Guangde

On behalf of all the authors, I would like to ask you to consider our manuscript entitled” A case of a perforated obturator hernia with a femoral abscess repaired using the pectineus muscle” for publication in BMC Surgery as a case report. This report highlights the repair of an obturator hernia with a femoral abscess by using the pectineus muscle, which is of special interest because it was effective as prevention of the recurrence of the abscess. The patient has provided permission to publish these features of her case, and the identity of the patient has been protected.

We feel that the findings described in this case report will be of special interest to the readers of BMC Surgery.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission.

Sincerely,

Naoki Kubo M.D
Department of Surgery, Azumi General Hospital, 3207-1, Ikeda, Ikeda-cho, Kitaazumi-gun,
Nagano 399-8695, Japan
Phone number: +81 261 62 3166
Fax number: +81 261 78 2248
E-mail: nkazumihp@yahoo.co.jp